Mini MOSS 2 Debrief Minutes

July 7, 2004

ELEMENT ATTENDANCE:

ESDIS SSDO, EDOS, ASDC DAAC, TES SIPS, GES DAAC, MLS SIPS, HIRDLS SIPS, OMI SIPS

NEXT TELECON:  As-needed basis

DIAL-IN NUMBER:

USA Toll Free Number: 866-581-4760

USA Toll Number: +1-210-280-9695 (international)

PASSCODE: 574686
MEETING HIGHLIGHTS:

All debrief presentations can be obtained from the Aura Testing website: http://jupiter02.gsfc.nasa.gov/AuraTesting/ under Mini-MOSS-2 link. Only highlights, changes, or clarifications will be presented below.

SSDO:

Karen mentioned that she was trying to setup a special telecon after the debrief today with EDOS and TES SIPS to discuss the handling of flushing out short PDSs. Karen was unable to reach Ron before the meeting to confirm the meeting so she stated that she would get a hold of Ron and try to schedule something for tomorrow (7/8/04). After the telecon, Karen sent an email to the TES SIPS proposing to have the telecon at 2:30pm EST on 7/8.

Karen thanked everyone for their excellent support and wished them well for the mission. 

EDOS:

Nancy presented the slides for EDOS. She indicated that the first five RBDs to OMI NRTS were unsuccessful due to a ssh configuration issue at ONRTS. Once corrected, all files were transmitted without any problems. EDOS sent out all products to the users.

GES DAAC:

Ed presented the GES DAAC slides via telecon. He reported that they had problems with g0drg05 mid-way during the test and the problems were not related to this test. They had to turn off pollEDOS for ~4 hours. Aura ingest was held up because their EDOS delivered data arrive in the same polling directory. GES DAAC received and distributed all that were expected except delivery of one EDS for APID 1631 due to early distribution problem. GES DAAC can re-delivery the file to HIRDLS at any time via a User Services request. The extra EMOS files (~750) received during the early part of the test were corrected and EMOS (Dan Linebarger) informed Karen that they wrote a DR.

MLS:
Audrey presented the slides via telecon. She reported that the processing of the V1.43 L1 PGE, using the Mini MOSS 2 data relayed from the SIPS, was completed successfully at the SCF. MLS received, processed, and delivered all that were expected except receipt of one PDS for APID 1738 which she indicated that it was not needed for processing.

OMI SIPS:

Marghi presented the slides via telecon. She indicated that she was very happy with the outcome of the test. They received, processed, and distributed all that were expected. They did not produce L2s due to the MT2 data was unsuitable for L2 PGE processing and they did not have any simulated data on the system.

HIRDLS:

Brooke presented the slides via telecon.  HIRDLS received all granules except an EDS for APID 1631 (consistent with GES DAAC report). Due to a software issue, HIRDLS2 data were not produced during this test. However, the previous Mini MOSS 2 HIRDLS2 products were sent to the GES DAAC and were successful ingested and archived. Karen asked Brooke to keep her posted on the L2 software issue.
ASDC DAAC: 

Sue presented the slides via telecon. Overall, ASDC successfully ingested L0 and DPREP data and distributed them to the TES SIPS via scp; and successfully ingested the TES L1B and L2 products on July 3 and 4. Two L2 products failed ingest due to ShortName mismatch in the PDR vs metadata. Andy thought that the problem was corrected. Andy is still trying to find out what happened.

TES SIPS: 
Andy presented the slides via telecon. He reported that TES received all products that were expected. L2 products were delivered to ASDC on Friday July 2nd (AM) and L1Bs on July 3rd (PM). Andy mentioned that they integrated all new PGEs between Mini MOSS 1 and Mini MOSS 2.

EMOS: Slides posted on the web site.
ODPS: Slides posted on the web site. 

FDS: Not required to participate. Products retrieved from GES DAAC archive for the test.
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